Author’s response to reviews

Title: Nurse-identified Patient Care and Health Services Research Priorities in the United Arab Emirates

Authors:

Nabeel Al-Yateem (nalyateem@sharjah.ac.ae)
Muna Al-Tamimi (mtamimi@sharjah.ac.ae)
Maria Brenner (Brennerm@tcd.ie)
Hanan Altawil (tawil@sharjah.ac.ae)
Alaa Ahmad (alaa@seha.ae)
Sharon Brownie (sharo.brownie@aku.edu)
Shameran Slewa-Younan (S.Younan@westernsydney.edu.au)

Version: 4 Date: 05 Jan 2019

Author’s response to reviews:

Comment 1: Authors Contributions

Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Response: Author contributions section modified as requested.

Comment 2: Figure 1

Please state in the cover letter whether the map depicted in figure 1 is your own or taken from another source.

Response: I confirm that this figure is prepared by myself and not taken from another source.

Comment 3:

Clean Manuscript
Please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please also ensure that all figures, tables, and supplementary/additional files are cited in the text.

Response: ALL comments, highlights, etc were removed as requested.